Author's response to reviews

Title: Effectiveness of Continuous Glucose Monitoring during Diabetic Pregnancy (GlucoMOMS trial); a randomised controlled trial.

Authors:

Daphne N. Voormolen (d.p.vanmunster-2@umcutrecht.nl)
J. Hans DeVries (j.h.devries@amc.uva.nl)
Arie Franx (a.franx-2@umcutrecht.nl)
Ben W.J. Mol (b.w.mol@amc.uva.nl)
Inge M. Evers (im.evers@meandermc.nl)

Version: 2 Date: 29 October 2012

Author's response to reviews:

Please hereby find the revise version of the manuscript on the GlucoMOMS trial. In the Methods section, a clarification was added on the approval of the ethics committee.

Kind regards,
Daphne Voormolen